UCSF UC San Francisco Previously Published Works

Title

High-Density SNP Screening of the Major Histocompatibility Complex in Systemic Lupus Erythematosus Demonstrates Strong Evidence for Independent Susceptibility Regions

Permalink

https://escholarship.org/uc/item/8jf557n9

Journal PLOS Genetics, 5(10)

ISSN 1553-7390

Authors

Barcellos, Lisa F May, Suzanne L Ramsay, Patricia P <u>et al.</u>

Publication Date 2009-10-01

DOI

10.1371/journal.pgen.1000696

Peer reviewed

High-Density SNP Screening of the Major Histocompatibility Complex in Systemic Lupus Erythematosus Demonstrates Strong Evidence for Independent Susceptibility Regions

Lisa F. Barcellos¹, Suzanne L. May¹, Patricia P. Ramsay¹, Hong L. Quach¹, Julie A. Lane², Joanne Nititham³, Janelle A. Noble², Kimberly E. Taylor³, Diana L. Quach¹, Sharon A. Chung³, Jennifer A. Kelly⁴, Kathy L. Moser⁴, Timothy W. Behrens⁵, Michael F. Seldin⁶, Glenys Thomson⁷, John B. Harley⁴, Patrick M. Gaffney⁴, Lindsey A. Criswell³*

1 Division of Epidemiology, School of Public Health, University of California Berkeley, Berkeley, California, United States of America, 2 Children's Hospital of Oakland Research Institute, Oakland, California, United States of America, 3 Rosalind Russell Medical Research Center for Arthritis, University of California San Francisco, San Francisco, California, United States of America, 4 Oklahoma Medical Research Foundation, Oklahoma City, Oklahoma, United States of America, 5 Immunology Diagnostics and Biomarkers, Genentech, South San Francisco, California, United States of America, 6 University of California Davis, California, United States of America, 7 Department of Integrative Biology, University of California Berkeley, Berkeley, California, United States of America

Abstract

A substantial genetic contribution to systemic lupus erythematosus (SLE) risk is conferred by major histocompatibility complex (MHC) gene(s) on chromosome 6p21. Previous studies in SLE have lacked statistical power and genetic resolution to fully define MHC influences. We characterized 1,610 Caucasian SLE cases and 1,470 parents for 1,974 MHC SNPs, the highly polymorphic *HLA-DRB1* locus, and a panel of ancestry informative markers. Single-marker analyses revealed strong signals for SNPs within several MHC regions, as well as with *HLA-DRB1* (global $p = 9.99 \times 10^{-16}$). The most strongly associated *DRB1* alleles were: *0301 (odds ratio, OR = 2.21, $p = 2.53 \times 10^{-12}$), *1401 (OR = 0.50, p = 0.0002), and *1501 (OR = 1.39, p = 0.0032). The MHC region SNP demonstrating the strongest evidence of association with SLE was rs3117103, with OR = 2.44 and $p = 2.80 \times 10^{-13}$. Conditional haplotype and stepwise logistic regression analyses identified strong evidence for association between SLE and the extended class I, class I, class II, class II, and the extended class II MHC regions. Sequential removal of SLE-associated *DRB1* haplotypes revealed independent effects due to variation within *OR2H2* (extended class I, rs362521, p = 0.006), *CREBL1* (class III, rs8283, p = 0.01), and *DQB2* (class II, rs7769979, p = 0.003, and rs10947345, p = 0.0004). Further, conditional haplotype analyses demonstrated that variation within *MICB* (class I, rs3828903, p = 0.006) also contributes to SLE risk independent of *HLA-DRB1*0301*. Our results for the first time delineate with high resolution several MHC regions with independent contributions to SLE risk. We provide a list of candidate variants based on biologic and functional considerations that may be causally related to SLE risk and warrant further investigation.

Citation: Barcellos LF, May SL, Ramsay PP, Quach HL, Lane JA, et al. (2009) High-Density SNP Screening of the Major Histocompatibility Complex in Systemic Lupus Erythematosus Demonstrates Strong Evidence for Independent Susceptibility Regions. PLoS Genet 5(10): e1000696. doi:10.1371/journal.pgen.1000696

Editor: Derry C. Roopenian, The Jackson Laboratory, United States of America

Received May 26, 2009; Accepted September 23, 2009; Published October 23, 2009

Copyright: © 2009 Barcellos et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: Support for this study was provided by the NIH (AR052300, AR02175, AR22804, AR62277, AR42460, Al24717, Al53747, Al31584, DE15223, RR20143, Al62629, AR48940, AR19084, AR043274, Al063274, Al40076, AR052125), research grants from the US Department of Veterans Affairs, American College of Rheumatology, Alliance for Lupus Research, Rheuminations, the Lupus Foundation of Minnesota, and the Mary Kirkland Center for Lupus Research. This study was performed in part in the General Clinical Research Center, Moffitt Hospital, University of California San Francisco, with funds provided by the National Center for Research Resources, 5 M01 RR-00079, US Public Health Service. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: lindsey.criswell@ucsf.edu

Introduction

Systemic lupus erythematosus (SLE) is the prototypic systemic autoimmune disease characterized by autoantibody production and involvement of multiple organ systems. Although the etiology of SLE remains unknown, several lines of evidence underscore the importance of genetic factors, including a high sibling risk ratio ($\lambda_s = 8-29$), familial clustering, where approximately 10 to 12% of SLE patients have an affected first-degree relative, and higher concordance rates in monozygotic twins (24–69%) relative to dizygotic twins and non-twin siblings (2–9%) [1,2,3]. Similar to many other autoimmune diseases, genes within the major histocompatibility complex (MHC) on the short arm of chromosome 6p21.3 exhibit strong association with the risk for SLE.

The MHC is a gene-dense region of the human genome spanning approximately 4.5 Mb and known to encode more than 180 expressed genes [4,5]. Forty percent of the expressed loci have functions related to immune activation and response. The class III region, with more than 55 expressed loci, is the most dense subregion of the MHC and the entire human genome [6].

Historically, interest in the MHC region for SLE has focused on the highly polymorphic HLA class I and II genes that encode

Author Summary

Systemic lupus erythematosus (SLE) is an autoimmune disease characterized by autoantibody production and involvement of multiple organ systems. Although the cause of SLE remains unknown, several lines of evidence underscore the importance of genetic factors. As is true for most autoimmune diseases, a substantial genetic contribution to disease risk is conferred by major histocompatibility complex (MHC) gene(s) on chromosome 6. This region of the genome contains a large number of genes that participate in the immune response. However, the full contribution of this genomic region to SLE risk has not yet been defined. In the current study we characterize a large number of SLE patients and family members for approximately 2,000 MHC region variants to identify the specific genes that influence disease risk. Our results, for the first time, implicate four different MHC regions in SLE risk. We provide a list of candidate variants based on biologic and functional considerations that may be causally related to SLE risk and warrant further investigation.

membrane glycoproteins that present peptides for recognition by T lymphocytes, as well as genes within the HLA class III region, particularly the tumor necrosis factor and complement component C4 gene loci. Indeed, inherited deficiency of complement genes, particularly C4A (null) alleles, has long been recognized as strong, albeit rare, genetic risk factors for SLE [7]. Work by Graham and colleagues [8,9] examining \sim 50 microsatellite markers across the MHC region highlighted the importance of HLA class II haplotypes involving the HLA-DRB1 and -DQB1 loci, particularly those corresponding to serologic types HLA-DR2 and DR3 (DRB1*1501-DQB1*0602 and DRB1*0301-DQB1*0201, respectively). More recently, genome wide association scans in SLE also underscore the importance of the MHC [10,11,12]. However, strong linkage disequilibrium (LD) between particular alleles within the MHC has interfered with disease variant identification and previous studies have not been able to distinguish between associated MHC variants. Long-range, extended or 'ancestral' haplotypes, sometimes spanning greater than 2 Mb, have been observed [8,13]. Recent studies in European-derived populations have examined the distribution of LD across the MHC and have suggested that SNPs can help dissect causal variation within this region [14,15].

A recent analysis of 314 UK SLE families examining 68 SNPs across the HLA class II and III regions and *HLA-DRB1* suggested two distinct association signals centered on *DRB1*0301* and the T allele of rs419788 in intron 6 of the class III region gene *SKIV2L* [16]. Of interest, the class III signal appears to exclude the *TNF* - *308* promoter polymorphism, which has been a focus of prior work [17]. Herein we extend this work by characterizing a much larger collection of SLE cases (and parents) for 1,974 MHC genetic markers plus the *HLA-DRB1* locus. We also characterized SLE cases for a set of ancestry informative markers (AIMs) to identify population outliers and assess the possible impact of substructure within the European population on our genetic association results. Further, we utilized several analytic strategies to determine whether multiple distinct alleles or haplotypes contribute to SLE risk.

Results/Discussion

Multiple Regions within the MHC Influence SLE Risk

Table	I. Demo	graphic	and	clinical	characteris	stics o	of	study
subjects	5.							

Case Series (# SLE cases)*	# trio families ($#$ parents)
UCSF (925)	325 (650)
OMRF (429)	170 (340)
UMN (256)	240 (480)
Total # SLE cases: 1,610	Total # parents: 1,470
Female, %	92
Age at SLE diagnosis, mean (standard deviation)	33.1 (13.2)
History of renal involvement [†] , %	32
dsDNA autoantibody production [‡] , %	54
Ethnicity (based on ancestry informative marker data)	1,577 of 1,610 successful
≥80% European ancestry	1,518 (96.3%)
≥90% European ancestry	1,392 (88.3%)
\geq 90% Northern European ancestry (estimated for those with \geq 90% European ancestry)	1,130 (71.1%)

* UCSF = University of California, San Francisco; OMRF = Oklahoma Medical Research Foundation, UMN = University of Minnesota.

[†]renal involvement based on presence of American College of Rheumatology renal criterion [35] and/or renal biopsy findings consistent with lupus nephritis. [‡]dsDNA = double strand DNA.

doi:10.1371/journal.pgen.1000696.t001

families included in this study. Clinical characteristics of the SLE cases are consistent with those reported for Caucasian patients, generally [18].

Table 2 summarizes association results for HLA-DRB1 alleles in SLE cases and non-transmitted control alleles from parents of the trio families [19]. Due to the highly polymorphic nature of this locus and prior evidence suggesting that multiple HLA-DRB1 alleles influence SLE risk, we employed a relative predispositional effects (RPE) analysis [20] (see Methods). Results indicated that three specific HLA-DRB1 alleles were strongly associated risk: DRB1*0301 (odds ratio, OR = 2.21, with SLE $p = 2.53 \times 10^{-12}$, *1401 (OR = 0.50, p = 0.0002) and *1501 (OR = 1.39, p = 0.0032). The *DRB1*0801* allele, which has been shown in previous work to be associated with SLE risk [8] was not significantly associated with SLE in the current RPE analysis, after accounting for testing of multiple DRB1 alleles (full results shown in Table S1).

Figure 1 displays single marker association results for the 1,974 MHC SNPs (out of 2,360) passing quality control filters (see Methods) in SLE cases compared to controls (additional results are shown in Table S2). We observed strong association signals across a broad region encompassing the HLA class I, III and II regions, with weaker evidence of association in the extended class I and II regions. The strongest evidence of association was observed at the multiallelic *DRB1* locus (global $p = 9.99 \times 10^{-16}$ and ORs as shown in Table 2). The MHC region SNP demonstrating the strongest evidence of association with SLE was rs3117103 (bp 32,457,535), with OR = 2.44 and $p = 2.80 \times 10^{-13}$. This SNP is flanking C6orf10 at a distance of 9.8 kb.

Given the extensive LD across this region, we utilized conditional analyses to identify association signals that were not due to rs3117103. Similarly, we repeated these analyses conditioning instead on the *HLA-DRB1* locus, including the *DRB1*0301*, **1401* and **1501* alleles. The rs number and map position for all SNPs associated with SLE using the conditional haplotype method (CHM) (total = 171 SNPs with p<0.01) are

Table 1 summarizes characteristics of the 1,610 Caucasian SLE cases and available parents (n = 1,470) from the complete trio

Table 2. Association of HLA-DRB1 alleles with SLE risk based on relative predispositional effects (RPE) analysis of 1,522 SLE cases and 693 controls.

HLA-DRB1 allele	Case allele frequency	Control allele frequency	P-value [†]	Odds Ratio (95% C.I.) [‡]
*0301	0.206	0.110	2.53×10 ⁻¹²	2.21 (1.82, 2.68)
*1401	0.015	0.035	0.0002	0.50 (0.33, 0.75)
*1501	0.165	0.141	0.0032	1.39 (1.15, 1.66)
*0801	0.031	0.023	0.0293	1.59 (1.05, 2.39)

[†]P-values based on RPE analysis and therefore account for the other associated *DRB1* alleles (see Methods). [‡]Reference group for odds ratio calculations included all *DRB1* alleles except *0301, *0801, *1401, and *1501.

doi:10.1371/journal.pgen.1000696.t002

provided in Table S3. Following these conditional analyses, several analytic approaches were utilized to further define genetic variants with the most compelling evidence of association with SLE risk, as shown in Figure 2.

Figure 3 displays initial association results for the MHC region variants (n = 11, including *HLA-DRB1*) that were shown to be associated with SLE risk based on conditional haplotype and stepwise logistic regression analyses, with p<0.001 (See Figure 2 and Methods). In addition to the *HLA-DRB1* locus, there were several variants with evidence of association with SLE, including one variant in the extended class II region, five variants in HLA class II (including *HLA-DRB1*), one variant in HLA class III, two in HLA class I, and two in the extended class I region.

Figure 4 shows an LD plot representing the correlation (r^2) among *HLA-DRB1* and the aforementioned 10 associated SNPs that met our significance threshold of p<0.001 based on conditional analyses (see Methods and Table S4). There was very

little evidence of LD among these 10 SNPs and *HLA-DRB1* with the exception of SNPs within the *MICB* ($r^2 = 0.58$), *C6orf10* ($r^2 = 0.53$) and *HLA-DQB2* loci ($r^2 = 0.35$) and strong LD between *HLA-DRB1* and the *C6orf10* locus ($r^2 = 0.77$).

To further evaluate the evidence supporting association of these 10 variants with SLE we performed transmission disequilibrium testing [21,22] among 650 complete trio families. These analyses were consistent with the prior results, including strongest evidence of association for rs3117103 (OR = 2.44, p = 1.58×10^{-14}). Additional details are shown in Table S5.

We also evaluated these 10 SNPs among a more homogeneous subset of cases (and non-transmitted controls) estimated by AIMs to have $\geq 90\%$ Northern European ancestry. These results also supported our main analyses, with the strongest evidence of association for rs3117103 (OR = 2.49, p = 1.62×10^{-13}) (see Table S5 for additional details). Similarly, results for the top four *DRB1* risk alleles did not change when analyses were



Figure 1. Association of 544 MHC region SNPs (out of 1,974) and *HLA-DRB1* with SLE. Shown are results for variants with p < 0.01 among 1,484 SLE cases and 650 controls. UNPHASED v.3.0.10 was used for 1,974 SNPs and SAS v. 9.1.3 for *HLA-DRB1* global p-value. Odds ratio shown for *HLA-DRB1* corresponds to *0301 allele. doi:10.1371/journal.pgen.1000696.g001



Figure 2. Summary of analytical plan. Summary of analytical plan for defining MHC region genetic variants with independent contributions to SLE risk among 1,484 SLE cases and 650 controls. doi:10.1371/journal.pgen.1000696.g002

repeated among the Northern European subset of cases and controls (data not shown).

A distinguishing feature of the MHC region is the extensive, long-range LD, which has been observed particularly among European-derived populations [15] and is haplotype specific [23]. Thus, to further assess the independence of the 11 variants (including DRB1) and determine whether one or more extended haplotypes are strongly associated with SLE risk, we estimated haplotypes involving these variants and evaluated the evidence of association with SLE (see Methods). When case and control chromosomes were considered together, a total of 171 unique haplotypes were observed. When MHC haplotypes were compared between SLE cases and controls, evidence of association was revealed for two common haplotypes. The haplotype with the strongest evidence of association with SLE contains the DRB1*0301 allele (15% transmitted vs. 6% non-transmitted haplotypes, OR = 2.63, $p = 8.32 \times 10^{-15}$), which represents ${\sim}70\%$ of the DRB1*0301 haplotypes. The other associated haplotype contains the *1501 allele (9% transmitted vs. 6% nontransmitted haplotypes, OR = 1.54, p = 0.0025) (data not shown).

Finally, we considered our data both a) excluding and b) conditioning on haplotypes containing the SLE-associated *DRB1* alleles. More specifically, we compared MHC SNP allele frequencies among cases vs. controls after sequentially removing haplotypes containing *0301, *1501, and then *1401 (see Table 3 and Table S6). After removing haplotypes containing these alleles, we observed no evidence for independent association of rs3117103 with SLE (p>0.30; see Table 3 and Table S6). However, four MHC region variants remained significantly associated with SLE,

including rs362521 (OR2H2, extended class I region; p = 0.0060), rs8283 (CREBL1, class III; p = 0.011), rs7769979 and rs10947345 (DQB2, class II; p = 0.0030 and 0.0004, respectively). Of interest, two HLA class II region SNPs at the DOB2 locus remained associated with SLE even in the absence of DRB1 risk alleles. This is also the first study to support a role in SLE for variants within the extended MHC class I region, near OR2H2. Next, the CHM was used to investigate associations with the top 10 MHC region SNPs conditioned on haplotypes containing the SLE-associated HLA-DRB1 alleles (see Table 4 and Table S6). When conditioning on DRB1*0301 haplotypes, we observed significant association with SLE (p = 0.0059) for a SNP at the *MICB* locus (HLA class I). In contrast, conditioning on DRB1*1501 or DRB1*1401 revealed no evidence for other MHC effects on SLE risk, however limited power was present for conditional analyses involving DRB1*1401 given the infrequency of that allele in this dataset. Finally, logistic regression analyses involving the aforementioned five SNPs and HLA-DRB1 revealed evidence of independent association with SLE for all variants (p-values = 0.003 to $< 10^{-6}$) with the exception of one of the DQB2 SNPs (rs7769979) (data not shown).

In summary, we have characterized a large set of SLE cases and parents of European ancestry for the highly polymorphic HLA-DRB1 locus and ~2,000 SNPs across 4.9 Mb of the MHC. We have employed methods to ensure that our main results are not due to confounding by population admixture or major substructure within the European population. Our results support the existence of multiple, independent association signals across this region as opposed to a single primary association signal (with other effects explained by LD). Independent associations within



Figure 3. Summary of MHC region variants with evidence of independent association with SLE. Summary of MHC region variants (n = 11, including *HLA-DRB1*) with evidence of independent association with SLE (p<0.001) based on conditional haplotype and stepwise logistic regression analyses. Number in parentheses indicates number of SNPs at that locus. See Figure 2 for details of overall analytical strategy. P-values from original univariate analyses are shown on the y-axis. doi:10.1371/journal.pgen.1000696.g003

extended class I and both class II and III regions were present, even after accounting for *HLA-DRB1* effects. However, we cannot exclude the possibility that multiple associated variants within MHC haplotypes function together to influence SLE risk through co-regulation or other mechanisms.

An important limitation of the current analysis is the lack of comprehensive coverage in the MHC region. For example, the MHC Panel Set of markers was not designed to directly assess rare variants or multi-allelic markers, such as insertion-deletion or copy number variants. In addition, we have not characterized these families for other classical HLA genes (besides *HLA-DRB1*), and in fact our results implicate a role for other HLA class II region genes. Thus, we cannot determine which, if any, of the associated variants we have identified represent underlying causal variants, versus proxies for causal variants as a result of LD. In order to begin to address this important question we identified SNPs and other variants tagged by the independently associated SNPs and summarized their known function, as well as what is known about these variants themselves (see Table S7).

One of the novel loci highlighted in the current study is *OR2H2*, within the extended class I region, which encodes olfactory receptor family 2, subfamily H, number 2. The olfactory receptor proteins are G-protein-coupled receptors that share structure with many neurotransmitter and hormone receptors [24]. To our knowledge, this gene has not been implicated in risk of SLE or other human autoimmune diseases. Our top SNP in this region tags three additional SNPs in a single locus, gamma-aminobutyric acid (GABA) B receptor, 1 (*GABBR1*, see Table S7), which is also a neurotransmitter receptor [25]. Of relevance to this finding, recent chromosome 6 sequencing efforts have now fully characterized the

extended MHC in humans [26], spanning a total of 7.6 Mb and comprised of five subregions that include the original 'classical' MHC [27]. Of the 421 loci in this region, a total of 252 (60%) are classified as expressed genes. Although our study included 305 SNPs within the extended class I MHC and 247 within the extended class II MHC, more work is needed to interrogate the entire extended MHC region.

Our top SNP within the class III region was located in the 3' untranslated region of *CREBL1*, which encodes c-AMP responsive element binding protein-like 1 [28]. Of interest, we did not identify additional variants tagged by our top SNP in this region (see Table S7). Unfortunately, our MHC marker set did not include SNPs within or near the *C4* locus, which has been previously associated with SLE. In particular, copy number variation at this locus has been strongly implicated in SLE risk. However, the two markers flanking the *C4* locus did not reveal evidence for association with SLE in the current study (rs389512 and rs1009382, p>0.15). Given the density of genes within the class III region and previous work suggesting a contribution of class III variants to SLE risk, further study of this region is clearly warranted, including direct assessment of *C4* variation.

We also demonstrated association of SLE with a variant near the 5' end of the MHC class I chain-related B (*MICB*) gene, which encodes a heavily glycosylated protein that serves as a ligand for the *NKG2D* type 2 receptor [29]. Binding of this ligand activates the cytolytic response of natural killer cells. This finding is also of interest given previous evidence of association of the closely related *MICA* gene to SLE [30], although results have been contradictory [31]. The *MICB* gene has not been previously implicated in SLE risk. All other variants tagged by our top SNP in this region are within or very near the *MICB* locus (see Table S7).



Figure 4. Linkage disequilibrium (LD) plot for *HLA-DRB1* **and 10 MHC region SNPs.** LD plot showing correlation (r^2) for *HLA-DRB1* and 10 MHC region SNPs demonstrating evidence of independent association with SLE. A total of 2,607 individuals with >90% European ancestry (maximum unrelated subset of all cases and trios) were used for analysis. Similar results were obtained when only the SLE cases or SLE trio family founders were used to generate the LD plots (data not shown). *HLA-DRB1* was coded as a biallelic marker (*DRB1*0301* vs. others) for analysis in Haploview based on results obtained for LD analyses of this locus as a multilocus marker using UNPHASED (see Methods). See Table 3 for other gene names. doi:10.1371/journal.pgen.1000696.g004

		All Haps [†]		DRB1*0301 removed		DRB1*0301, *1501 removed		<i>DRB1*0301</i> , <i>*1501</i> , <i>*1401</i> removed	
SNP	Gene	P-value	OR	P-value	OR	P-value	OR	P-value	OR
rs1536215	TRIM27	0.0037	0.72	0.0806	0.81	0.0627	0.78	0.0684	0.78
rs362521	OR2H2	0.0021	0.62	0.0065	0.64	0.0081	0.61	0.0060	0.60
rs3828903	МІСВ	3.20×10 ⁻⁷	0.67	0.0024	0.78	0.0759	0.86	0.0965	0.86
rs2246626	МІСВ	3.45×10^{-4}	0.77	0.2570	0.91	0.7664	1.04	0.6680	1.04
rs8283	CREBL1	7.67×10 ⁻⁷	0.63	0.0013	0.74	0.0117	0.78	0.0111	0.78
rs7746019	C6orf10	7.85×10 ⁻¹²	1.77	0.1075	1.20	0.0244	1.30	0.0224	1.31
rs3117103	C6orf10	2.42×10 ⁻¹⁶	2.35	0.3961	1.47	0.3701	1.52	0.3699	1.50
rs7769979	DQB2	5.64×10 ⁻¹⁴	0.58	1.90×10 ⁻⁶	0.69	0.0013	0.75	0.0030	0.76
rs10947345	DQB2	6.96×10 ⁻⁴	1.29	1.21×10^{-7}	1.53	0.0002	1.42	0.0004	1.40
rs383711	HSD17B8	0.0024	0.65	0.0308	0.73	0.0949	0.77	0.1082	0.77

Table 3. Results from relative predispositional effects analyses of top MHC SNPs in SLE cases and controls.

OR = odds ratio.

[†]Haplotypes assigned in SLE cases and all family members using PHASE v. 2.1.2. Non-transmitted haplotypes were used as control group. Haplotypes containing SLEassociated *HLA-DRB1* alleles were sequentially removed (see Methods).

doi:10.1371/journal.pgen.1000696.t003

Table 4. Results from conditional haplotype method analyses of top MHC SNPs in SLE cases and controls.

SNP		DRB1*0301 only		DRB1*1501 only		DRB1*1401 only	
	Gene	P-value	OR	P-value	OR	P-value	OR
rs1536215	TRIM27	0.16	0.48	0.7701	0.91	1.000	0.89
rs362521	OR2H2	0.43	0.68	0.4213	0.72	NA	NA
rs3828903	МІСВ	0.0059	0.46	0.1216	0.61	0.3422	0.47
rs2246626	МІСВ	0.0615	0.59	0.4478	0.80	0.3614	0.48
rs8283	CREBL1	0.0122	0.32	1.0000	1.02	0.2220	NA
rs7746019	C6orf10	0.0189	1.94	0.6853	2.05	0.7403	1.40
rs3117103	C6orf10	0.0189	1.94	1.0000	1.36	NA	NA
rs7769979	DQB2	0.0544	0.54	0.2374	0.70	0.2451	0.30
rs10947345	DQB2	0.8908	1.06	0.2536	1.37	0.2451	3.35
rs383711	HSD17B8	0.2428	0.43	0.2013	0.59	0.5988	0.43

OR = odds ratio

[†]Haplotypes assigned in SLE cases and all family members using PHASE v. 2.1.2. Non-transmitted haplotypes were used as control group.

doi:10.1371/journal.pgen.1000696.t004

Lastly, two HLA class II region SNPs at the DQB2 locus remained associated with SLE even in the absence of DRB1 risk alleles, and these two SNPs tag variants of at least four additional class II loci (see Table S7). However, our study lacked power to detect additional rare DRB1 associations (beyond *0301, *1501 and *1401), thus it is possible that additional DRB1 effects explain, at least in part, the DQB2 association observed in the current study.

We also sought to determine how our results compare to those reported by Fernando, et al., based on their analysis of 68 SNPs in the HLA class II and III regions, plus *HLA-DRB1*, in 314 UK SLE families [16]. Two distinct association signals centered on *DRB1*0301* and the T allele of rs419788 in intron 6 of the class III region gene *SKIV2L* were observed. Five SNPs within this gene were included in our marker panel, including their top SNP, rs419788. Four of the five SNPs were associated with SLE with p<0.01 in our single marker analyses, however, only two of these SNPs (but not rs419788) met the significance threshold (p<0.01) for our CHM analyses. Neither of these two SNPs were among the group of ten SNPs selected for further study based on conditional haplotype based analyses or stepwise logistic regression.

We have not examined other, non-European populations in this study and this represents a very important focus for future work given the increased burden of SLE among those groups. Further, comparison of MHC association results across major ethnic groups, where LD patterns are substantially different, or the analysis of specific disease subtypes may improve our ability to localize causal variants within this region.

Methods

Ethics statement

Written consent was obtained from all study participants and ethical approval for this study was obtained from the University of California, San Francisco Committee on Human Research.

Subjects

SLE trio families and cases were derived from three independent case series, including the UCSF Lupus Genetics Project collection, the Lupus Genetics Studies and Lupus Family Registry and Repository (http://lupus.omrf.org/) at the Oklahoma Medical Research Foundation (OMRF) and the University of Minnesota SLE collection. SLE cases from trio families (n = 735) were a subset of the total set of SLE cases (n = 1,610), as shown in Table 1. Details of recruitment and enrollment procedures have been reported previously [32,33,34]. All subjects were self-reported Caucasian and all SLE cases met the American College of Rheumatology classification criteria for SLE [35,36]. Non-transmitted *HLA-DRB1* and MHC SNP alleles (described below) from SLE trio family parents were used as 'controls' for all comparisons with SLE cases, as previously described [19].

DNA samples

DNA was extracted from whole blood using the Gentra Puregene Whole Blood Kit. DNA was obtained from buccal cells and extracted using the Gentra Puregene Mouse Tail Kit. A total of 50 ng of buccal cell or 25 ng of blood DNA (when necessary) was whole genome amplified (WGA) using the Sigma WGA2 Kit prior to genotyping. WGA samples were column purified using the Sigma Genelute PCR Clean-up Kit. DNA was also extracted from DNA Genotek Oragene saliva sample collection kits according to the manufacturer's protocol. Of the total 3,080 individual DNA samples used for genotyping in this study, 69% were derived from whole blood, 25% from buccal cells, and 6% from saliva; 72% of all samples (including both blood and buccal) were WGA prior to genotyping.

Genotyping methods

HLA-DRB1. Genotyping for the classical *HLA-DRB1* locus was performed using a PCR-Sequence-Specific Oligonucleotide Probe (SSOP) methodology [37], also known as "linear array" technology. Briefly, this technology employs biotin-labeled amplification products, generated from the polymorphic exons of the HLA genes, in a hybridization reaction with a series of unlabelled oligonucleotides. These oligonucleotides, which correspond to variable sequence motifs in the genes, are immobilized onto a nylon backed membrane. Hybridization is visualized colorimetrically, and signal intensity is automatically quantitated with a flatbed scanner and in-house software. Genotype calling was achieved using SCORE, HLA genotyping software developed by Dr. Wolfgang Helmberg (www.ncbi.gov). The DRB1 HiRes assay specifically amplifies only *DRB1* alleles, thus removing the confounding signals from other *DRB* loci that

can be found in commercial assays. A total of 100–200 ng was used, depending on DNA source.

MHC SNP genotyping. All DNA samples were genotyped for MHC region SNPs using the Combined MHC genotyping panel (2,360 SNPs) from Illumina. This panel spans 4.9 Mb across the MHC region with \sim 2 kb/SNP spacing and utilizes the robust Golden-Gate technology [38]. The Combined MHC Panel Set is comprised of both the Exon-Centric and Mapping panels. The Exon-Centric Panel consists of 1,228 SNPs which are located in or near coding regions (within 10 kb of exons). The Mapping Panel consists of 1,293 SNPs. Genotyping was performed in a 96-well format utilizing the 1,536 Sentrix Array Matrix (×2). A total of 250 ng to 1 ug was used for each Illumina assay, depending on source of DNA.

Ancestry informative markers (AIMs). A total of 384 SNP markers selected for both Continental and European ancestry informativeness, including differentiation between Northern vs. Southern European origin were genotyped for all SLE cases [39,40] (see Table S4). Genotypes were derived using a custom Illumina Golden Gate 384 SNP panel (~65% of SLE cases) or the Illumina HumHap 550 BeadChip (~35% of cases). A total of 22 samples were genotyped on both platforms for comparison (see QC methods below)

Quality control analysis for DNA samples and **genotypes.** Stringent quality control criteria were applied to all samples and genotype data (HLA-DRB1, MHC region SNPs, and AIMs). Sample success rate for HLA-DRB1 was 97.9%. Mendelian inconsistencies for DRB1 were identified in five SLE trio families. These individuals (n = 15) were excluded from further analyses. A per-sample genotype rate threshold of 80% was used for MHC SNP genotyping (total n = 173 samples removed); a per-SNP genotype rate threshold of 90% was used (total n = 337 SNPs removed). Additional criteria including assessment of SNP minor allele frequency (<0.01), Hardy-Weinberg equilibrium (p < 0.001), and both Mendelian (>0.01) and gender inconsistencies identified six additional families (n = 18 individuals) and 49 additional SNPs for exclusion from analyses. Analysis of replicate SNPs (n = 146 SNPs occurring on both Illumina Mapping and Exoncentric panels) demonstrated very strong evidence for agreement; 99.7% of all genotypes for all samples compared were the same. Replicate sample comparisons within and across DNA genotyping plates, as well as between different sample sources (blood vs. saliva or blood vs. buccal) also demonstrated very strong evidence for agreement (data not shown). A total of 2,873 samples (665 trios and 878 SLE cases without parents) and 1,974 MHC SNPs passed our stringent quality control criteria. The mean distance between MHC SNPs was 2.08 kb (range: 0.005-71.05 kb). The total MHC SNP coverage spanned 4,913.28 kb. Similar quality control criteria were used for all AIMs genotype data. A total of 27 SNPs were removed from analysis, leaving a final dataset comprised of genotypes from 357 SNPs (Table S4). A total of 22 samples were genotyped using both platforms for comparison demonstrating very high agreement (data not shown). Five additional SLE cases and three additional trio families were removed from further analyses due to missing AIM data. Results of AIM genotyping and analysis among all SLE cases who reported Caucasian ancestry revealed 51 SLE cases with an estimated European ancestral contribution <80% and these individuals (plus their parents, total n = 75) were excluded from further analysis. Two datasets were created for analysis of HLA-DRB1 and MHC SNP loci based on European ancestry: >80% European ancestry (650 trio families and 834 SLE cases; total n = 2,784 individuals); >90% European ancestry (614 trio families and 765 SLE cases; total n = 2,607 individuals). The first of these datasets (>80% European ancestry) was used for initial univariate and conditional haplotype analyses (see below). The second dataset (>90% European ancestry) was used for subsequent analyses. A third dataset comprised of an even more genetically homogeneous subgroup of cases (n = 1,379) who were estimated based on AIMs to have at least 90% Northern (vs. Southern) European ancestry was also used to examine the final MHC SNP variants identified in the study.

Analysis of HLA-DRB1. Global testing for heterogeneity at the *DRB1* locus was performed using SAS v. 9.1.3 (Cary, NC). A method to reveal the relative predispositional effects (RPEs) (predisposing, protective, or neutral) of the *HLA-DRB1* alleles was also used, as previously described [20,41]. When a disease is associated with two or more alleles at a locus, the RPE method identifies the associations sequentially according to their strength; thus the problem that a strong association with one allele can create misleading deviations in the frequencies of other alleles is alleviated.

Univariate analysis of MHC variants. Univariate analysis of MHC region SNPs was performed using the log likelihood ratio test implemented in the UNPHASED program v.3.0.10 [42]. Transmission disequilibrium testing (TDT) [21] analyses of individual markers were performed using the PLINK program [22].

Conditional methods of analysis. The conditional haplotype method (CHM) [43,44], was used to initially distinguish primary from secondary MHC associations. The CHM is a test for homogeneity of relative allele frequencies in cases and controls (or transmitted and non-transmitted alleles in families) at a test locus on haplotypes identical for alleles at another locus. Specifically, the CHM tests the null hypothesis of equality in the distribution of transmitted (T) and nontransmitted (NT) marker haplotypes identical at one variant but different at another closely linked variant. If there is heterogeneity between the T and NT (case-control) groups in the distribution of two marker haplotypes identical at a predisposing marker (variant A) but different at a putative predisposing marker at another site (variant B), then this is evidence that variant A does not entirely explain disease predisposition and that variant B itself, or another marker in LD with variant B, is influencing the relative association of variant A, and therefore disease risk. We performed initial analyses of all MHC SNPs in the SLE cases and non-transmitted controls, conditioning on the top SNP and each DRB1 susceptibility allele. SNPs associated with SLE at p<0.01 based on initial univariate analyses were included in these analyses (n = 544 SNPs; see Figure 2). All haplotype counts were determined using UNPHASED v.3.0.10 [42]. In addition, the CHM was used for analysis of the 11 final MHC variants. Here, full haplotypes including *HLA-DRB1* were assigned using PHASE v.2.1.1, as described below. The Fisher's exact test in SAS was used to calculate OR, 95% CI and p-values for each analysis.

We also used the WHAP program [45] version 2.09 (http://pngu.mgh.harvard.edu/~purcell/whap/) to further evaluate the variants identified using the CHM to have some evidence of independent association with SLE susceptibility (p<0.01; see above and Figure 1). A per-block WHAP method was utilized. More specifically, within each haplotype block, we performed a conditional analysis starting with *HLA-DRB1*, which was modeled as a multi-allelic marker with values *DRB1*0301*, *DRB1*1401*, *DRB1*1501*, or other. Starting with *DRB1*, we tested each candidate marker for an independent effect; in particular this uses a likelihood-ratio test to determine the significance (condi-

tional p-value) of the difference between an "alternate" model with 2-locus haplotypes - composed of *DRB1* plus each new candidate - versus the initial 1-marker (*DRB1* only) "null" model. As long as the most significant locus out of the list of candidates has conditional p<0.01, we add this best marker to the list of independent markers (now *DRB1* plus one SNP), and proceed to test all remaining SNPs as part of 3-marker haplotypes - again composed of the current list plus each new candidate marker compared to our 2-locus model; and so on with larger haplotypes as appropriate. The algorithm completes when no remaining candidate SNP has p<0.01 conditional on the current list of significant markers. Haplotype block structure for WHAP analysis was defined using solid spline (http://www.biostat.wustl.edu/ genetics/geneticssoft/manuals/haploview/haploview_doc.pdf).

The Tagger program [46] was used to select MHC SNPs identified from CHM analyses (p<0.01 as described above) for further analysis using stepwise logistic regression modeling, and also for retrieving tagged SNPs from CEPH Hap Map data as previously described (see below). Stepwise logistic regression analyses were performed using STATA v.9.2 (College Station, Texas). Forward stepwise was performed using thresholds of p<0.01 to add a variable and p≥0.0101 to remove a variable from our model. For the stepwise logistic regression analyses, MHC SNPs were coded for presence/absence of the minor allele. *HLA-DRB1* was coded according to genotype for SLE associated alleles (0301, 1501, and 1401), with all other alleles as the reference group. Logistic regression analysis including the final top MHC SNPs and *HLA-DRB1* was performed using a similar approach.

Imputation of missing MHC SNP data. Missing genotypes for cases and non-transmitted controls were inferred using default parameters in BEAGLE v2.1.3 [47]. The entire dataset served as the reference group. Briefly, four haplotype pairs were sampled per individual for each iteration of the phasing algorithm. Ten iterations of the phasing algorithm were performed. The threshold scale and threshold shift, which are used to control the number of haplotype clusters in the graphical model for the phased data, were set to 4.0 and 0.2, respectively. The maximum number of consecutive markers considered for determining whether to merge haplotype clusters was 500. The number of missing genotypes imputed in this study was small, at 1.47% of all study genotypes (across 1,974 SNPs and 2,784 individuals, or ~ 5.5 million genotypes). We used the resulting phased data file, which gives the most likely genotypes for each individual, as input for stepwise logistic regression analyses of cases and controls described above.

Linkage disequilibrium analyses. Haploview (v.4.1) [48] was used to define LD among the final group of MHC SNP variants using all cases and founders from SLE families. Similarly, for the multi-allelic *HLA-DRB1* locus, all pairwise measures of LD with MHC region SNPs were determined using UNPHASED (-LD option) [42]. Due to the restriction of Haploview to analyze and display measures of LD for biallelic markers, *HLA-DRB1* was recoded. We chose *DRB1*03001* vs. other alleles for this analysis based on results obtained for LD analyses of this locus as a multilocus marker using UNPHASED.

Haplotype assignment and analysis. PHASE v.2.1.1 was used to assign extended multilocus haplotypes in all SLE cases and parents comprised of *HLA-DRB1* and our top 10 MHC SNPs. Default settings for number of iterations (n = 100), thinning interval (n = 1) and burn-in (n = 100) were used. RPE analysis and tests of heterogeneity to compare MHC SNP effects on *HLA-DRB1* haplotypes were performed using SAS. OR, 95% CI and p-values are reported.

Identification of tagged variants. LD data calculated by HapMap for the CEPH population were downloaded directly from the HapMap Genome Browser (http://hapmap.org/ cgi-perl/gbrowse) for phase I and II SNPs (release #23a, NCBI Build 36, June 2008) and phase III SNPs (draft release #1, Sept 2008). For each study SNP, LD was assessed for all SNPs within 500 kb flanking each side of the SNP. In addition, for two of our study SNPs not present in NCBI Build 36, CEPH genotypes were downloaded directly from HapMap and LD between each of these SNPs and other SNPs within 500 kb in NCBI Build 35 was assessed using the Tagger algorithm as implemented in Haploview (v 4.1) by force including each SNP and force excluding all others. For each release and method, SNPs typed in the CEPH were considered tagged by our study SNPs if they showed $r^2 \ge 0.80$. SNPs shown to be tagged, but already typed and assessed in our study were not included as tagged variants. For study SNPs not typed in both phases, LD between these SNPs and SNPs typed only in the alternate phase could not be assessed. For study SNPs not included in NCBI Build 36, LD could only be assessed between these and SNPs typed in phase I and phase II that were also present in NCBI Build 35. Annotation for each study SNP and tagged SNP including nearest gene, SNP function, and heterozygosity was taken from the UCSC Genome Browser (http://genome.ucsc.edu/).

Supporting Information

Table S1 Relative predispositional effects (RPE) analysis of*HLA-DRB1* alleles among 1,522 SLE cases versus 693 controls.

Found at: doi:10.1371/journal.pgen.1000696.s001 (0.09 MB DOC)

Table S2Association results for 1,974 SNPs among 1,484 SLEcases and 650 controls.

Found at: doi:10.1371/journal.pgen.1000696.s002 (3.70 MB DOC)

Table S3 List of MHC SNPs (n = 171) associated with SLE with p < 0.01 based on conditional haplotype method among 1,484 SLE cases and 650 controls.

Found at: doi:10.1371/journal.pgen.1000696.s003 (0.16 MB DOC)

Table S4 Ancestry informative markers (total n = 357) genotyped to classify SLE cases (n = 1,570) according to continental ancestry (n = 112 SNPs) and European substructure (n = 246SNPs).

Found at: doi:10.1371/journal.pgen.1000696.s004 (0.39 MB DOC)

Table S5 Results from analyses of independently associated MHC variants in SLE families and Northern European individuals (see Figure 3 and Figure 4).

Found at: doi:10.1371/journal.pgen.1000696.s005 (0.04 MB DOC)

Table S6 Results from relative predispositional effects (RPE) (A) and CHM analyses (B) of top MHC SNPs in SLE cases and controls. (Same as Table 3 and Table 4 with 95% confidence intervals included.)

Found at: doi:10.1371/journal.pgen.1000696.s006 (0.08 MB DOC)

Table S7 Details for top associated study MHC SNPs plus variants tagged by these SNPs (see Figure 3, Figure 4, Table 3, Table 4).

Found at: doi:10.1371/journal.pgen.1000696.s007 (0.15 MB DOC)

Acknowledgements

The authors wish to thank Gary Artim for programming assistance, Kenny Beckman for laboratory consultation, the SLE patients and families, referring physicians, and the UCSF DNA Banking facility.

References

- Alarcon-Segovia D, Alarcon-Riquelme ME, Cardiel MH, Caeiro F, Massardo L, et al. (2005) Familial aggregation of systemic lupus erythematosus, rheumatoid arthritis, and other autoimmune diseases in 1,177 lupus patients from the GLADEL cohort. Arthritis Rheum 52: 1138–1147.
- Block SR, Winfield JB, Lockshin MD, D'Angelo WA, Christian CL (1975) Studies of twins with systemic lupus erythematosus. Am J Med 59: 533–552.
- Deapen D, Escalante A, Weinrib L, Horwitz D, Bachman B, et al. (1992) A revised estimate of twin concordance in systemic lupus erythematosus. Arthritis Rheum 35: 311–318.
- (1999) Complete sequence and gene map of a human major histocompatibility complex. The MHC sequencing consortium. Nature 401: 921–923.
- Stewart CA, Horton R, Allcock RJ, Ashurst JL, Atrazhev AM, et al. (2004) Complete MHC Haplotype Sequencing for Common Disease Gene Mapping. Genome Res 14: 1176–1187.
- Xie T, Rowen L, Aguado B, Ahearn ME, Madan A, et al. (2003) Analysis of the gene-dense major histocompatibility complex class III region and its comparison to mouse. Genome Res 13: 2621–2636.
- Walport MJ (2002) Complement and systemic lupus erythematosus. Arthritis Res 4 Suppl 3: S279–293.
- Graham RR, Ortmann WA, Langefeld CD, Jawaheer D, Selby SA, et al. (2002) Visualizing human leukocyte antigen class II risk haplotypes in human systemic lupus erythematosus. Am J Hum Genet 71: 543–553.
- Graham RR, Ortmann W, Rodine P, Espe K, Langefeld C, et al. (2007) Specific combinations of HLA-DR2 and DR3 class II haplotypes contribute graded risk for disease susceptibility and autoantibodies in human SLE. Eur J Hum Genet 15: 823–830.
- Harley JB, Alarcon-Riquelme ME, Criswell LA, Jacob CO, Kimberly RP, et al. (2008) Genome-wide association scan in women with systemic lupus erythematosus identifies susceptibility variants in ITGAM, PXK, KIAA1542 and other loci. Nat Genet 40: 204–210.
- Hom G, Graham RR, Modrek B, Taylor KE, Ortmann W, et al. (2008) Association of systemic lupus erythematosus with C8orf13-BLK and ITGAM-ITGAX. N Engl J Med 358: 900–909.
- Graham RR, Cotsapas C, Davies L, Hackett R, Lessard CJ, et al. (2008) Genetic variants near TNFAIP3 on 6q23 are associated with systemic lupus erythematosus. Nat Genet 40: 1059–1061.
- Aly TA, Baschal EE, Jahromi MM, Fernando MS, Babu SR, et al. (2008) Analysis of single nucleotide polymorphisms identifies major type 1A diabetes locus telomeric of the major histocompatibility complex. Diabetes 57: 770–776.
- Walsh EC, Mather KA, Schaffner SF, Farwell L, Daly MJ, et al. (2003) An integrated haplotype map of the human major histocompatibility complex. Am J Hum Genet 73: 580–590.
- Miretti MM, Walsh EC, Ke X, Delgado M, Griffiths M, et al. (2005) A highresolution linkage-disequilibrium map of the human major histocompatibility complex and first generation of tag single-nucleotide polymorphisms. Am J Hum Genet 76: 634–646.
- Fernando MM, Stevens CR, Sabeti PC, Walsh EC, McWhinnie AJ, et al. (2007) Identification of two independent risk factors for lupus within the MHC in United Kingdom families. PLoS Genet 3: e192. doi:10.1371/journal. pgen.0030192.
- Lee YH, Harley JB, Nath SK (2006) Meta-analysis of TNF-alpha promoter -308 A/G polymorphism and SLE susceptibility. Eur J Hum Genet 14: 364–371.
- Alarcon GS, McGwin G, Jr., Petri M, Reveille JD, Ramsey-Goldman R, et al. (2002) Baseline characteristics of a multiethnic lupus cohort: PROFILE. Lupus 11: 95–101.
- Thomson G (1995) Mapping disease genes: family-based association studies. Am J Hum Genet 57: 487–498.
- Payami H, Joe S, Farid NR, Stenszky V, Chan SH, et al. (1989) Relative predispositional effects (RPE's) of marker alleles with disease: HLA-DR alleles and Graves disease. Am J Hum Genet 45: 541–546.
- Spielman RS, Ewens WJ (1996) The TDT and other family-based tests for linkage disequilibrium and association. Am J Hum Genet 59: 983–989.
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MA, et al. (2007) PLINK: a tool set for whole-genome association and population-based linkage analyses. Am J Hum Genet 81: 559–575.
- Blomhoff A, Olsson M, Johansson S, Akselsen HE, Pociot F, et al. (2006) Linkage disequilibrium and haplotype blocks in the MHC vary in an HLA haplotype specific manner assessed mainly by DRB1*03 and DRB1*04 haplotypes. Genes Immun 7: 130–140.
- Fan W, Liu YC, Parimoo S, Weissman SM (1995) Olfactory receptor-like genes are located in the human major histocompatibility complex. Genomics 27: 119–123.

Author Contributions

Conceived and designed the experiments: LFB LAC. Performed the experiments: SLM HLQ JAL JAN DLQ. Analyzed the data: LFB PPR JN KET SAC MFS GT LAC. Contributed reagents/materials/analysis tools: LFB JAN KET JAK KLM TWB JBH PMG LAC. Wrote the paper: LFB KET LAC.

- Grifa A, Totaro A, Rommens JM, Carella M, Roetto A, et al. (1998) GABA (gamma-amino-butyric acid) neurotransmission: identification and fine mapping of the human GABAB receptor gene. Biochem Biophys Res Commun 250: 240–245.
- Mungall AJ, Palmer SA, Sims SK, Edwards CA, Ashurst JL, et al. (2003) The DNA sequence and analysis of human chromosome 6. Nature 425: 805–811.
- Horton R, Wilming L, Rand V, Lovering RC, Bruford EA, et al. (2004) Gene map of the extended human MHC. Nat Rev Genet 5: 889–889.
- Min J, Shukla H, Kozono H, Bronson SK, Weissman SM, et al. (1995) A novel Creb family gene telomeric of HLA-DRA in the HLA complex. Genomics 30: 149–156.
- Bahram S, Bresnahan M, Geraghty DE, Spies T (1994) A second lineage of mammalian major histocompatibility complex class I genes. Proc Natl Acad Sci U S A 91: 6259–6263.
- Gambelunghe G, Gerli R, Bocci EB, Del Sindaco P, Ghaderi M, et al. (2005) Contribution of MHC class I chain-related A (MICA) gene polymorphism to genetic susceptibility for systemic lupus erythematosus. Rheumatology (Oxford) 44: 287–292.
- Sanchez E, Torres B, Vilches JR, Lopez-Nevot MA, Ortego-Centeno N, et al. (2006) No primary association of MICA polymorphism with systemic lupus erythematosus. Rheumatology (Oxford) 45: 1096–1100.
- Thorburn CM, Prokunina-Olsson L, Sterba KA, Lum RF, Seldin MF, et al. (2007) Association of PDCD1 genetic variation with risk and clinical manifestations of systemic lupus erythematosus in a multiethnic cohort. Genes Immun 8: 279–287.
- Kaufman KM, Kelly JA, Herring BJ, Adler AJ, Glenn SB, et al. (2006) Evaluation of the genetic association of the PTPN22 R620W polymorphism in familial and sporadic systemic lupus erythematosus. Arthritis Rheum 54: 2533–2540.
- Graham RR, Langefeld CD, Gaffney PM, Ortmann WA, Selby SA, et al. (2001) Genetic linkage and transmission disequilibrium of marker haplotypes at chromosome 1q41 in human systemic lupus erythematosus. Arthritis Res 3: 299–305.
- Tan EM, Cohen AS, Fries JF, Masi AT, McShane DJ, et al. (1982) The 1982 revised criteria for the classification of systemic lupus erythematosus. Arthritis Rheum 25: 1271–1277.
- Hochberg MC (1997) Updating the American College of Rheumatology revised criteria for the classification of systemic lupus erythematosus [letter]. Arthritis Rheum 40: 1725.
- Bugawan TL, Apple R, Erlich HA (1994) A method for typing polymorphism at the HLA-A locus using PCR amplification and immobilized oligonucleotide probes. Tissue Antigens 44: 137–147.
- Oliphant A, Barker DL, Stuelpnagel JR, Chee MS (2002) BeadArray technology: enabling an accurate, cost-effective approach to high-throughput genotyping. Biotechniques Suppl: 56–58, 60–51.
- Tian C, Plenge RM, Ransom M, Lee A, Villoslada P, et al. (2008) Analysis and application of European genetic substructure using 300 K SNP information. PLoS Genet 4: e4. doi:10.1371/journal.pgen.0040004.
- Kosoy R, Nassir R, Tian C, White PA, Butler LM, et al. (2009) Ancestry informative marker sets for determining continental origin and admixture proportions in common populations in America. Hum Mutat 30: 69–78.
- proportions in common populations in America. Hum Mutat 30: 69–78.41. Thomson G, Barcellos LF, Valdes AM (2008) Searching for additional disease loci in a genomic region. Adv Genet 60: 253–292.
- Dudbridge F (2003) Pedigree disequilibrium tests for multilocus haplotypes. Genet Epidemiol 25: 115–121.
- Valdes AM, Thomson G (1997) Detecting disease-predisposing variants: the haplotype method. Am J Hum Genet 60: 703–716.
- Valdes AM, McWeeney S, Thomson G (1997) HLA class II DR-DQ amino acids and insulin-dependent diabetes mellitus: application of the haplotype method. Am J Hum Genet 60: 717–728.
- Purcell S, Daly MJ, Sham PC (2007) WHAP: haplotype-based association analysis. Bioinformatics 23: 255–256.
- de Bakker PI, Yelensky R, Pe'er I, Gabriel SB, Daly MJ, et al. (2005) Efficiency and power in genetic association studies. Nat Genet 37: 1217–1223.
- Browning SR, Browning BL (2007) Rapid and accurate haplotype phasing and missing-data inference for whole-genome association studies by use of localized haplotype clustering. Am J Hum Genet 81: 1084–1097.
- Barrett JC, Fry B, Maller J, Daly MJ (2005) Haploview: analysis and visualization of LD and haplotype maps. Bioinformatics 21: 263–265.